

### Water Resources Research

### **RESEARCH ARTICLE**

10.1029/2018WR022976

### **Special Section:**

Dynamics in Intensively Managed Landscapes: Water, Sediment, Nutrient, Carbon, and Ecohydrology

#### **Key Points:**

- The Murray-Darling Basin, which covers one million square kilometers, was developed to support irrigated agriculture from the early 1900s
- The secondary salinization that followed became a major concern, and mitigation strategies have since reduced salinity levels
- Evidence from long-term monitoring of macroinvertebrates suggests that salinity mitigation has been effective in reversing past impacts

### Correspondence to:

W. L. Paul, w.paul@latrobe.edu.au

### Citation:

Paul, W. L., Cook, R. A., Suter, P. J., Clarke, K. R., Shackleton, M. E., McInerney, P. J., & Hawking, J. H. (2018). Long-term monitoring of macroinvertebrate communities over 2,300 km of the Murray River reveals ecological signs of salinity mitigation against a backdrop of climate variability. *Water Resources Research*, *54*, 7004–7028. https://doi.org/10.1029/ 2018WR022976

Received 20 MAR 2018 Accepted 5 SEP 2018 Accepted article online 10 SEP 2018 Published online 27 SEP 2018

©2018. American Geophysical Union. All Rights Reserved.

### Long-Term Monitoring of Macroinvertebrate Communities Over 2,300 km of the Murray River Reveals Ecological Signs of Salinity Mitigation Against a Backdrop of Climate Variability

Warren L. Paul<sup>1</sup>, Robert A. Cook<sup>1</sup>, Phillip J. Suter<sup>1</sup>, K. Robert Clarke<sup>2</sup>, Michael E. Shackleton<sup>1</sup>, Paul J. McInerney<sup>1</sup>, and John H. Hawking<sup>1</sup>

<sup>1</sup>Centre for Freshwater Ecosystems, School of Life Sciences, La Trobe University, Wodonga, Victoria, Australia, <sup>2</sup>Plymouth Marine Laboratory, Plymouth, UK

Abstract We investigated the ecological effects of salinity mitigation strategies in the Murray-Darling Basin (MDB) using macroinvertebrate data collected over 2,300 km of the Murray River between 1980 and 2012. The MDB covers  $1 \times 10^{6}$  km<sup>2</sup> and includes both temperate and semiarid climate zones. It was extensively developed to support irrigated agriculture in the early to mid-1900s, and the secondary salinization that followed has become a major concern. During 1975-1985 daily salinity levels, measured as electrical conductivity above the Murray River off-take points for South Australia's major urban water supplies, were above 800 µS/cm for 40% of the time, necessitating mitigation strategies that have reduced the average salinity by about 150 µS/cm since monitoring began. The MDB has also experienced several major floods and droughts during this time, and surface temperatures in the MDB have increased by 0.8 °C since 1910, mostly in the last 50 years. We hypothesized that (1) taxa richness would increase in response to floods; (2) community structure would shift toward tolerant, opportunistic taxa in response to warming; and (3) geographical ranges of species would change in response to shifting stream isotherms and reducing salinity. Our hypotheses were supported, although increases in water temperature appeared to be due principally to the 1997–2009 Millennium drought. Importantly, against a backdrop of significant climate variability, we believe we have distinguished a change in community structure along a salinity gradient and that changes over the 33 years can in part be attributed to mitigation strategies.

**Plain Language Summary** Groundwater in the Murray-Darling Basin is naturally salty as a result of the low rainfall and high evaporation that has concentrated salt borne by the rain over tens of thousands of years. The introduction and expansion of irrigated agriculture since the early 1900s rose water tables over large areas and increased the discharge of saline groundwater into the Murray River. The increasing salinity in the lower Murray River raised environmental and public health concerns in the 1970s, and a program of salinity mitigation was implemented. Using long-term data, we have documented changes occurring along 2,300 km of the Murray River over a 33-year period. Aquatic macroinvertebrates—which includes insects, crustaceans, mussels, snails, and worms that spend all or part of their life in water—have increased or decreased in abundance in response to natural factors such as floods and droughts and also to changes brought about by the construction of weirs and the side effects of agriculture on the pH and salinity of river water. Importantly, there are ecological signs that salinity mitigation strategies have been effective in reversing past impacts.

### 1. Introduction

Secondary salinization of rivers has in the past led to the collapse of irrigated agricultural systems and the civilizations that depended on them (Jacobsen & Adams, 1958; Pillsbury, 1981), and even with today's scientific and technological proficiency it presents a global and growing threat to agricultural production and aquatic ecosystems (Cañedo-Argüelles et al., 2013; Pillsbury, 1981). This is especially the case in inland arid and semiarid regions where large amounts of salt are concentrated in soils and subsequently leached into rivers through rising water tables caused by irrigation (Cañedo-Argüelles et al., 2013; Smedema & Shiati, 2002). More than half of the world's large river systems are either moderately or severely affected by dams built primarily to support irrigated agriculture (Nilsson et al., 2005), and irrigation has been reported to be responsible for the salinization of rivers such as the Amu Darya and Syr Darya in Central Asia, the Breede

River in South Africa, the Ebro River in Spain, the Great Menderes River in Turkey, and the Colorado River in the United States (Cañedo-Argüelles et al., 2013; Pillsbury, 1981). Some countries are now investing hundreds of millions of dollars to mitigate and control the salinity of rivers (Murray-Darling Basin Authority, MDBA, 2015; Pillsbury, 1981). Australia is one such example, where the Murray River, one of the large river systems identified by Nilsson et al. (2005) as severely impacted, has been the focus of a salinity mitigation program for over 30 years (MDBMC, 2015; Schrobback et al., 2008).

Development throughout the Murray-Darling Basin (MDB) in the twentieth century saw widespread environmental degradation of its rivers and wetlands from regulation for transport, irrigated agriculture, and hydroelectricity generation, and the growth of cities and towns. In the 1920s and 30s, 14 weirs, 13 with locks, were constructed along the Murray River to aid navigation and irrigation, and to provide reliable water supplies to towns, and the first stage of the Hume Dam was completed in 1936 (present capacity 3,040 GL). In the lower reaches of the Murray River below the confluence with the Darling River most of the weir pools of each storage extend to the storage upstream, making the river more like a series of lakes, particularly when river flows are low. Irrigation, originally conceived in the late nineteenth century as a means of reducing climatic risk and encouraging closer settlement of the hinterland, expanded rapidly throughout the MDB for 60 years until about 1980 (Musgrave, 2008). Dam construction also continued through this period to meet the needs of irrigators and the need for hydroelectricity generation to secure the nation's power supply, despite warnings about the economics of irrigation and hydropower and the failure of closer settlement schemes (Crase, 2008; Musgrave, 2008). The MDB is now referred to as Australia's food bowl, accounting for one third of the national food supply, and while irrigated agriculture occupies less than 2% of the MDB's  $1 \times 10^{6}$  km<sup>2</sup> (14% of Australia's land surface), it accounts for 40% of the MDB's agricultural output and about two thirds of Australia's total agricultural water use (Australian Bureau of Statistics, 2008).

The ensuing environmental problems now known to be associated with irrigation and dams (Nilsson et al., 2005) were compounded by two factors in the MDB. First, the extreme variability in rainfall and inflows requires that storages be more than twice the size of the world's average, or six times the size of the European average, to achieve the same level of supply security (Musgrave, 2008; Smith, 1998); a corollary being that the disruption to the natural flow regime is likely to be greater in Australia (McMahon & Finlayson, 1995; Musgrave, 2008). The long-term annual inflow to the MDB is a modest 31,600 GL, with a range of 6,700 to 117,900 GL (Water Act 2007 - Basin Plan, 2012). Regulation has reduced the natural average annual flow at the mouth of the Murray River by 62%, from 12,498 to 4,712 GL (Maheshwari et al., 1995). The natural spring flood pulse has been virtually eliminated, and the timing of the seasonal peak flow has shifted at Albury, below the Hume Dam, from winter-spring to summer-autumn, with little change evident further downstream in the timing of peak flows (Maheshwari et al., 1995). Second, in common with many semiarid areas of the world, groundwater in the MDB is highly saline. Rain-borne salt has accumulated in sediments over tens of thousands of years of relative aridity (Evans et al., 1990; Herczeg et al., 2001), and by raising the water table irrigation and tree clearing have contributed to the salinization of large areas of land and the export of salt to rivers (Macumber, 1990). A time series analysis of chloride measurements for the Murray River, taken between 1938 and 1981 at the town of Morgan, upstream of the Murray River off-take points for South Australia's major urban water supplies, revealed that chloride levels were seasonal, with higher mean levels occurring between March and May when flows were lower, and that mean levels had increased exponentially by 0.97% per annum after adjusting for flow (Cunningham & Morton, 1983), or 51% over the 43-year period. Between 1975 and 1985 daily salinity levels, measured as electrical conductivity (EC), exceeded the World Health Organization's guideline for human consumption of 800 µS/cm for 40% of the time at Morgan (Blackmore et al., 2000). Salinity levels have remained above 800 µS/cm for prolonged periods during times of drought and low flow. Daily salinity levels were above 800 µS/cm at Morgan for 31 months between January 1965 and June 1968 (a period of drought and low flow), with levels peaking at 1,400 µS/cm in February 1968. In the early 1980s, during a low flow period, salinity at Morgan was above 800  $\mu$ S/cm for 18 consecutive months, with a peak of 1,400  $\mu$ S/cm in July 1982 (MDBMC, 2015).

Urban development created another set of problems for the MDB's rivers. The population of the MDB grew to two million, and urban sewerage and storm water drainage, which was contributing almost half of the nutrient load to the MDB's rivers during dry periods, was implicated as the main cause of algal blooms, one of the worst stretching over 1,000 km of the Darling River in 1991 (Smith, 1998).

From the 1980s and particularly in the 1990s the growing realization of the unsustainability of current practices led to ongoing policy reforms and initiatives to halt the growth in water diversions in the MDB, provide for environmental flows, and address the issues of rising salinity levels and algal blooms. Guided by the National Strategy for Ecologically Sustainable Development (McKay, 2008), a cap was placed on diversions for consumptive uses in the MDB in 1995, limiting diversions to the 1993–1994 level, and in 1997 water entitlements began to be converted from volumetric entitlements to share entitlements of the consumptive pool (McKay, 2008) to enable adaptive management of the resource into the future (Pagan, 2008). In 2002 the MDBMC established The Living Murray Initiative, and in 2004 The Living Murray Program began recovering water for environmental flows, as well as building management structures to enable flooding of wetlands and lakes (MDBA, 2016; MDBC, 2007). The Living Murray and other programs have so far recovered about 74% of the target of 2,750 GL annually (MDBA, 2017b), to achieve the long-term sustainable diversion limit of 10,870 GL/year (Water Act 2007 - Basin Plan, 2012). The MDBMC also initiated the Salinity and Drainage Strategy in 1988 to reduce salinity through a program of lowering groundwater levels and constructing a salt interception scheme in some irrigation districts, which to date has reduced salt loads by 400,000 t/year and achieved an average reduction in EC of 150 µS/cm at Morgan (MDBMC, 2015). In 1994 the MDBMC developed The Algal Management Strategy for the Murray-Darling Basin to reduce nutrient pollution in the MDB's rivers (MDBMC, 1994). Biomonitoring programs were established during this period to evaluate the effects of control measures, according to the requirements of the Murray-Darling Basin Agreement and the Basin Plan (Water Act 2007 - Basin Plan, 2012). One of the longest running programs is the River Murray Water Quality Monitoring Program (WQMP), which started in 1980, spans 2,300 km of the Murray River, and includes both water quality and macroinvertebrate monitoring (Lawrence & Paterson, 2005). While there are other long-term (Daufresne et al., 2007; Floury et al., 2013; Fruget et al., 2015; Jackson & Füreder, 2006) and large-river-scale (Angradi et al., 2009) macroinvertebrate monitoring programs that have been reported in the literature, this is to the best of our knowledge the longest running, large-river-scale monitoring program reported thus far.

Since monitoring started, there have been several major flooding events and intense drought periods that have impacted on flows in the Murray River. Most notable are the flood events in 1993 and 2010–2011 that caused widespread flooding along the river (MDBA, 2017a), the drought of 1982–1983 that contributed to the closure of the Murray mouth for the first time in recorded history (MDBA, 2017a), and the Millennium drought from 1997 to 2009, in which southeastern Australia experienced the most persistent rainfall deficit on record (CSIRO, 2012; MDBA, 2017a). In addition, surface temperatures in the MDB have increased by 0.8 °C since 1910 (Timbal et al., 2015), with the greatest increase in the last 50 years, and there is evidence of a drying trend that may be attributable to climate change (CSIRO, 2012; Neave et al., 2015). Indicative calculations for lowland streams by Isaak and Rieman (2013) suggest that a long-term warming rate of 0.1–0.2 °C/decade could produce a shift in stream isotherms of 1.3–25 km/decade, though the shift is likely to be much greater during times of drought (Mosley, 2015). The effects of restoration efforts on macroinvertebrate assemblages must therefore be assessed against a backdrop of a variable and changing climate.

Aquatic macroinvertebrates have been extensively studied in the field and laboratory, and the body of literature that now exists enables some predictions to be made with regard to how macroinvertebrate communities are structured along rivers, and how they would respond to changes in water quality, episodic floods and droughts, and gradual warming. The river continuum concept (Vannote et al., 1980) predicts that the structure and function of macroinvertebrate communities conform to continuous gradients in width, depth, and velocity of a river, as well as the type and location of bioenergetic inputs. Further, the continuous gradient in community structure can be disrupted or reset through regulation by dams (the serial discontinuity concept of Ward & Stanford, 1983), inputs from tributaries, or anthropogenic disturbances such as nutrient enrichment, organic pollution, and changes in land use (Vannote et al., 1980). Numerous studies have reported the effects on macroinvertebrate communities of anthropogenic disturbances such as pollution, land use and riparian alteration, and associated changes in water guality (Chessman, 1995; Chessman, 2003; Horrigan et al., 2005; Kefford et al., 2006, 2011; Quinn et al., 1997; Reynoldson et al., 1997; Rutherford & Kefford, 2005). Increasingly long-term biological monitoring programs using macroinvertebrates have been valuable for detecting long- and short-term patterns in the ecological health of rivers and identifying drivers of change (Bradley & Ormerod, 2001; Daufresne et al., 2007; Jackson & Füreder, 2006; Vaughan & Ormerod, 2012), including changes due to drought/flood disturbances and directional trends in macroinvertebrate communities that may be associated with global warming (Chessman, 2009; Daufresne et al., 2007; Floury et al., 2013; Fruget et al., 2001). Two of these studies (Daufresne et al., 2007; Fruget et al., 2001) indicate that the effects of hydroclimatic factors are generally consistent with the intermediate disturbance hypothesis (Connell, 1978), in that species richness is low during steady conditions of low flow and high temperature but increases during periods of flood as space is opened for colonization by less competitive individuals (Townsend & Scarsbrook, 1997). Extreme hydroclimatic events consistently produce shifts in community structure, including patterns of appearance/disappearance of some taxa that may be consistent with latitudinal/altitudinal shifts in geographic range due to climate change and a shift toward euryto-lerant and invasive taxa with gradual warming (Daufresne et al., 2007). There is also some evidence that community structures do not recover following these hydroclimatic events and that their sensitivity to such events might decrease over time (Daufresne et al., 2007). Where interventions have been made to improve water quality, long-term studies generally indicate that the effects were secondary to those of gradual warming (Daufresne et al., 2007; Fruget et al., 2007; Fruget et al., 2001, 2015). The general consistency of these findings suggests that similar patterns will be observed in the macroinvertebrate communities of the Murray River.

Several studies have specifically addressed the effects of salinity on aquatic macroinvertebrates, many in an Australian context, but only community level predictions can be made at this point in time. Kefford et al. (2006) measured the acute salinity tolerance of 110 macroinvertebrate taxa collected from the southern MDB, and Kefford et al. (2004) found that acute tolerance is a reasonable reflection of the maximum field distribution (the maximum salinity at which a taxon is recorded in the field), but these measures provide no indication of changes in species abundance that sometimes occur at low salinities (Rutherford & Kefford, 2005). Schäfer et al. (2011) developed a salinity index (SPEAR<sub>sal</sub>) based on a database of the physiological sensitivities to salinity of macroinvertebrate taxa from southeast Australia, but while it may be useful for signaling changes due to salinity the index only predicts that sensitive taxa are replaced by tolerant taxa as salinity increases. Species response curves would be more useful for making predictions regarding changes in salinity. Horrigan et al. (2005) developed salinity sensitivity scores (for selected families) from samples taken across Queensland streams and made predictions about the probability of occurrence of those families with increasing salinity based on taxon-specific response curves. However, we note that for taxa that were generally absent from the more saline waters in the lower reaches of the Murray River during the 1980s, such as the freshwater yabby Cherax destructor (Parasticidae) and some caddisflies (Hydroptila spp.), their predictions were conflicting for samples taken from edge and riffle habitats. Furthermore, without more information on the particular genera or species observed in Horrigan's study, it is difficult to generalize their results to those from the Murray River WQMP. At the community level, Kefford et al. (2011) found that species richness appears to peak at an EC between 300 and 490 µS/cm in samples from across southeast Australia and that there was a statistically significant decline at <99 and >1,500  $\mu$ S/cm. Overall, it is reasonable to predict that community composition will have changed with reducing salinity in the lower Murray River (below the junction with the Darling River) and that species richness may have increased.

The aim of the present study is to describe and explain the spatiotemporal patterns in macroinvertebrate communities of the Murray River, and more specifically to assess the evidence for ecological signs of improvements in salinity. We expect that the macroinvertebrate communities will exhibit a spatial trend in response to environmental gradients in water quality, temperature, and flow and that this pattern will change over time in response to floods, gradual warming, and improvements in water quality. More specifically, we hypothesize that (1) the number of taxa will increase in response to floods, in accordance with the intermediate disturbance hypothesis, (2) the community will exhibit a shift toward tolerant, opportunistic taxa in response to warming, and (3) geographical ranges of species will change over time in response to shifting stream isotherms (due principally to drought) and reducing salinity.

### 2. Methods

### 2.1. Study Sites

The Murray River rises near Mount Kosciuszko at an elevation of 1,430 m, draining the western side of the Australian Alps, and flows 2,500 km to the Southern Ocean, near Goolwa in South Australia. The Murray River flows through four geomorphologically distinct tracts that may be identified as the Headwaters, Riverine, Mallee, and Lower Murray tracts, the detailed characteristics of which have been described



**Figure 1.** Map of the Water Quality Monitoring Program sites, with site numbers and distances from source (in parentheses).

elsewhere (Eastburn, 1990; Thoms et al., 2000). In the headwaters it is a swift flowing river upstream of the Hume Dam, at an elevation of 150 m and about 300 km from its source. Below Hume Dam the Murray flows westward in a shallow channel across the riverine plains, then through a deeper channel that cuts through the red and brown clays of the semiarid Mallee Zone. At Morgan the river turns south and travels the final 315 km, passing through the Lower Murray tract and Lake Alexandrina before discharging into the Southern Ocean. Annual flow in the Murray River is highly variable, with an average of around 11,000 GL, and a range of between 2,500 GL in a dry year and 40,000 GL in a wet year. Six sites were selected along the Murray River from Jingellic in the montane region of the catchment to Woods Point near the Murray River mouth, covering a distance of over 2,300 river kilometers (Figure 1). Site 801 lies within the Headwaters tract; Site 804 in the Riverine tract; Sites 808, 811, and 812 in the Mallee tract; and Site 814 is in the Lower Murray. Each site was located close to existing MDBA water quality monitoring sites to allow a comparison between water quality data and biological data (Bennison et al., 1989). Mean annual rainfall in the alpine headwaters is approximately 2,000 mm and the monitoring sites span a range of climates from temperate conditions in the upper catchment, with mean annual rainfall of 810 mm at the first site in Jingellic, through to a semiarid zone in the midreaches, with annual rainfall of approximately 260 mm, and a return to more temperate conditions toward the Murray River mouth, with mean annual rainfall of approximately 400 mm.

### 2.2. Macroinvertebrate Monitoring

Macroinvertebrates were collected using artificial substrate samplers (ASSs) in winter (May–June) and summer (October–December; Bennison et al., 1989). The data set used in this manuscript covers the period 1980 through to 2012 (Paul et al., 2018). The ASSs (Bennison et al., 1989) were constructed from a section of black plastic Gutterguard (mesh size 10 mm), 180 mm high and 240 mm in diameter containing one and a quarter new, commercially available polypropylene, leno weave onion bags (total area  $1,000 \times 420$  mm = 0.42 m<sup>2</sup>) as substratum. The Gutterguard was formed into a basket; onion bags were placed inside with a clean river rock or half brick for ballast and fastened with nylon cord top and bottom. Ten ASSs were placed on the substrate within the photic zone at each site and retrieved after a 6-week colonization period. About five ASSs were retrieved for each year/season/site combination. Samples were sieved through a 250-µm mesh sieve, then animals were removed and preserved in 70% ethanol. Samples were sorted and identified to the lowest taxonomic level possible (typically genus or species) and counted. Exceptions to this were (1) Nematoda and Nemertea were identified to phylum; (2) Oligochaeta and Hirudinea were identified to class; and (3) Acarina were identified to order. The current database includes 543 taxa, but due to changes in taxonomy over time these had to be grouped according to the 388 taxa known prior to 1985 (Bennison et al., 1989). Prior to analysis, this set of 388 taxa was reduced to 105 taxa (mostly genera and families) that were regularly collected (>10% of samples from any site).

### 2.3. Environmental Variables

Water quality data were obtained from the MDBA's WQMP (see Lawrence & Paterson, 2005; Paul et al., 2018). The WQMP includes weekly measurements of pH, turbidity, EC, water temperature, total Kjeldahl nitrogen (TKN), and total phosphorus (TP). All water quality variables were measured using standard methods (see Mackay et al., 1988).

Flow, or data used to calculate flow, is measured either continuously or instantaneously at a network of 140 gauging stations throughout the Murray River system (MDBA, 2012). Daily flow data were obtained from the nearest upstream gauging station for the present analysis.

Daily meteorological measurements of air temperature and rainfall in proximity to the water quality monitoring sites were obtained from the Bureau of Meteorology (2012). All observations were averaged over 12 weeks for two seasonal periods referred to as winter and summer, where the midpoint of each period coincided with the date of deployment of the ASSs used to monitor the macroinvertebrate communities.

### 2.4. Data Analysis

### 2.4.1. Modeling the Spatiotemporal Patterns in the Environmental Variables

The objective of this part of the analysis was to describe the spatiotemporal patterns in the environmental variables. Both space (distance from source) and time were entered into models as continuous variables. Modeling the spatial component as a function of distance from source, rather than as a function of site (a categorical variable), will produce a more general and parsimonious description of the data. The mgcv package in R (R Development Core Team, 2013; Wood, 2006) was used to fit generalized additive models (GAMs). The flexibility of GAMs makes them suited to the task of extracting the important features of the data without the need for complex parametric models (Wood, 2006). An identity link function was used in all models except for the rainfall model, which used a log link, and error distributions were Gaussian, or in the case of right-skewed distributions, inverse Gaussian or Gamma. Space-time plots for each variable were examined as part of an exploratory analysis, and diagnostics within mgcv were used to check the adequacy of the error distribution and link function used in each of the models.

## 2.4.2. Modeling the Abundance and Richness of the Macroinvertebrate Community as a Function of Space, Time, and the Environmental Variables

GAMs were also used to model the richness and total abundance of the community as a function of space and time, then as a function of the environmental variables with space and/or time included to account for any remaining systematic patterns. For both richness and abundance, a negative binomial error distribution and log link was used. Abundance and richness are defined as the average number of individuals per ASS retrieved and the number of taxa for each year/season/site combination, respectively, using the reduced set of 105 taxa. The environmental variables included pH, turbidity, EC, TKN, TP, water temperature, air temperature, rainfall, and flow. Current velocity is more important in structuring communities than is volumetric flow, but velocity data were not available. The normalized flow at each site, achieved by mean centering and scaling by the standard deviation, was therefore used to approximate the temporal effects of velocity on macroinvertebrate communities.

## 2.4.3. Modeling the Macroinvertebrate Community Structure as a Function of Space, Time, and the Environmental Variables

An exploratory principal coordinates (PCOs) analysis was done using the cmdscale function in the vegan R package (Oksanen et al., 2013). PCOs were computed from a between sample Bray-Curtis dissimilarity matrix of the fourth root transformed multivariate species abundance data, with a correction for negative eigenvalues (Legendre & Anderson, 1999). A fourth-root transformation of the abundance data was done to emphasize the contribution of less abundant taxa before calculating Bray-Curtis dissimilarity coefficients (Clarke, Gorley, et al., 2014; Clarke, Tweedley, & Valesini, 2014). The PCO scores represent the state of the community at a point in time and space, and systematic patterns in the community data will usually be revealed in the first few PCO axes, where each nontrivial PCO axis reflects an underlying ecological gradient (Faith et al., 1987; Gauch, 1982; Gauch et al., 1977; Legendre & Legendre, 2012; ter Braak & Prentice, 1988). Diagnostic tools were used to help identify the number of potentially nontrivial PCO axes (Paul & Anderson, 2013); these included the broken-stick method (Frontier, 1976), bootstrapped eigenvalue-eigenvector method (Jackson, 1993), and the holistic and conditional random permutation methods (Paul & Anderson, 2013). These tools can only serve as a guide to the number of ecological meaningful PCO axes. It is well known that ordinations produce three kinds of axes (Gauch, 1982; Gauch et al., 1977)-structure axes, spurious polynomial axes, and noise axes—and while these diagnostic tools can help to focus attention on a few potentially important axes, further examination of patterns in space-time plots of individual PCO scores is required in choosing those axes thought to contain ecological meaningful information.

Distance-based redundancy analysis (dbRDA) was used to model the community data as a function of space and time, as well as the environmental variables, using the capscale function in the vegan R package (Oksanen et al., 2013). The multivariate species data, expressed as a set of PCOs (as discussed above), were modeled first as a function of space and time (Legendre & Anderson, 1999; Paul & Anderson, 2013), then as a function of the environmental variables, with space and time included to account for the remaining systematic variation. This enabled an assessment of the percentage of the variation explained by the environmental variables and the percentage explained by space and time after partialling out the environmental component. Permutation tests for the regression coefficients were based on 999 permutations of residuals under a reduced model (Anderson & Legendre, 1999). The assumption of exchangeability, which underpins the permutation tests used in dbRDA, was checked by plotting the residuals versus fitted values for the first few PCO axes. In dbRDA the fitted PCO axes are usually subjected to an eigenvalue decomposition to produce a constrained ordination diagram, but it is sometimes preferable to plot the individual fitted and observed PCO values in space-time plots in order to visualize the trajectories and the level of agreement between the observed and fitted values (Paul & Anderson, 2013); it is also preferable with large data sets for which 2-D ordination diagrams appear cluttered. Spearman correlations between each PCO axis and the environmental variables are provided with the space-time plots to assist with the interpretation of the underlying ecological gradients.

With regard to interpretation of the variance explained by ordination axes, Økland (1999) notes that 30–70% of the total inertia is due to polynomial distortion axes that arise from lack-of-fit of the data to the species response model implicit in any eigenvector ordination method; that is, the eigenvalue-to-total-inertia ratio underestimates the compositional variation in structure axes. He recommends that interpretation of ordination analyses be based on relative values of explained variation for different sets of explanatory variables, rather than on absolute values of the fraction of variance explained by a particular set of variables. For this reason, we have also provided the variation explained by environmental variables as a percentage of the variation accounted for by space and time.

#### 2.4.4. Modeling Taxa Response Curves Along Ordination Axes

Shade plots produced with PRIMER v7 software (Clarke & Gorley, 2015) were used to visualize the patterns in coherent groups of taxa over the first four PCO axes of the macroinvertebrate analysis (Clarke, Tweedley, & Valesini, 2014). At its simplest, a shade plot is a data matrix in which entries are represented by a linear gray scale from white (the taxon is absent) to black (the maximum data value in the matrix). For this purpose, those taxa that were present in fewer than 10 of the 269 samples were eliminated because they were considered to have insufficient occurrences to be capable of providing meaningful patterns. The fourth-root transformed abundances were averaged in groups of 16-17 samples at a time, within 16 slices of variable width taken along each of the PCO axes (the x axes). Note that it is important to average over approximately constant numbers of samples, especially for strongly transformed abundances, if species accumulation artifacts are to be avoided. Taxa are ordered within the shade plots to reveal the role they play in determining the ecological gradients represented by the PCO axes. The idea here is that the structure among samples, which determines the PCO axes, is inherently driven by the structure among species—namely, their intercorrelations. On the y axis of the shade plot, the taxa are therefore grouped using coherent species analysis (Somerfield & Clarke, 2013), which uses Type 3 Similarity Profile (SIMPROF) tests to identify groups of taxa with statistically different responses among groups, over all samples, and statistically indistinguishable (coherent) patterns within each group. The ordering of these groups and the taxa within each group is then optimized using the method of Clarke, Gorley, et al. (2014). As suggested by Somerfield and Clarke (2013), we investigated the effect that different significance levels (0.1%, 0.5%, and 5%) for the Type 3 SIMPROF tests have on the number of coherent groups identified. A significance level of 0.5%, was deemed to produce a reasonably stable set of 30 groups. The lower significance level of 0.1% produced no groups, which would suggest this was an overly stringent criterion given the obvious structure among samples, as indicated by PCO analysis. The higher significance level (5%) produced more singletons. Further, on changing the significance level from 0.5% to 5% 26 of the 30 groups remained the same, with the four remaining groups (groups D, K, T, and AC) being split into two or three groups of one or more taxa.

In keeping with the aim of data reduction, the (generalized) mean abundance for coherent groups of taxa were then modeled as a function of the PCO axes. Modeling species response curves along individual ordination axes using generalized linear or additive models has been done previously (Lepš & Šmilauer, 2003; ter Braak, 1985; ter Braak & Šmilauer, 2002), but here we model coherent group average abundances as a function of the first four PCO axes in a multiple regression, rather than separately for each axis. In effect, this is analogous to the principal component solution of the factor model (Johnson & Wichern, 2002), where a few (nontrivial) principal components are taken to represent the latent common factors in a factor analysis. The response curve models used the generalized mean (to exponent ¼) of the taxa abundances for each coherent group, and the contributions of the first four PCOs were quantified using the deviance explained

PAUL ET AL.

by the individual smooth (PCO) terms and overall explained deviance, as well as the null deviance as a measure of the total variation (corrected for the mean) in the generalized mean abundance. The latter two statistics were used to rank the coherent groups in order of importance, while the explained deviance for individual terms was used as a measure of the *loading* of each PCO on the coherent group abundances, analogous to the estimated factor loadings in a factor analysis (Johnson & Wichern, 2002). GAMs with a log link and negative binomial error distribution were used for all coherent groups. As suggested by Oksanen and Minchin (2002) for modeling species distribution curves, a maximum effective degrees of freedom of 6 was used in all GAMs (by setting the basis dimension). This, however, is only an upper limit that is used to improve computational efficiency. The actual degrees of freedom, which was generally much lower, is controlled by the degree of penalization selected during fitting by generalized cross validation (Wood, 2006). Residual diagnostics within mgcv were used to check the adequacy of the error distribution and link function, as well as the choice of basis dimension, for each of the models.

# 2.4.5. Counterfactual Analysis to Elucidate How Environmental Changes Have Influenced Groups of Taxa

To predict the change in coherent group abundances that would occur under new environmental conditions, the PCO multiple regression model was first used to predict the PCO scores resulting from a change in the environmental variables, then the response curve models were used to predict abundances of coherent groups of taxa. This enabled analysis of counterfactual questions such as, *What would the macroinvertebrate community have been had the water temperature not warmed, or salinity mitigation strategies not been implemented*? See Pearl (2000, 2014) for further information on counterfactual analysis. Until a more complete causal model is developed, this analysis accounts only for the direct effects of environmental variables. It does not, for example, account for the indirect effect of temperature on macroinvertebrates that is mediated by water quality.

### 3. Results

### 3.1. Spatiotemporal Patterns in the Environmental Variables

All of the environmental variables, except rainfall, displayed a seasonal pattern. The fitted GAMs in Figure 2 were deseasonalized to highlight the longer-term patterns, though it is important to note that seasonal variation can be as ecologically important as long-term trends in means. All environmental variables except rainfall and ambient air temperature exhibited statistically significant distance-by-time interactions at the 0.05 significance level; that is, the spatial pattern varied over time.

Rainfall was highest in the upper catchment and lowest within the midreaches, with an increase again toward the Murray mouth (Figure 2f). The spatial pattern in ambient air temperature was the opposite, with higher temperatures in the midreaches and lower temperatures at either end of the river (Figure 2g). Rainfall was depressed and air temperature elevated during the droughts of 1982–83 and 1997–2009, whereas the reverse pattern occurred in the mid-1990s and 2010s, corresponding to floods in the Murray River.

The water quality variables—EC, turbidity, TKN, TP, water temperature, and pH—generally displayed an increasing trend downstream (Figures 2a–2d). Longitudinal patterns in flow were less clear except that highest flow rates were experienced at the lower end of the river system (Figure 2e). The magnitude of the spatial trends varied over time. EC declined in the mid-Murray sites from an average of about 400 to 200  $\mu$ S/cm (Sites 808–812, Figure 2b) and remained low at about 40  $\mu$ S/cm at the upper sites, 801 and 804. EC at Site 814 decreased from 800 to 400  $\mu$ S/cm and was particularly higher than at other sites during periods of drought in the early 1980s and mid-2000s. Turbidity was highly variable in the middle and lower reaches of the Murray River (Figure 2b), particularly in the sites downstream of the Darling River and Lake Victoria (Sites 811–814) but exhibited a gradual decline over the study period from an average of 60 to 40 NTU. Turbidity was generally less than 10 NTU at Site 801. TKN averages ranged between 0.2 and 0.8 mg N/L, and TP between 0.02 and 0.15 mg P/L. TKN and TP showed similar temporal trends to turbidity (graphs not shown). The pH at Site 801 diverged from other sites in the early 1980s (Figure 2a), decreasing from about 7.3 in the early 1980s to 6.6 in 2005. The water temperature (Figure 2d) increased by about 2.5 °C over the 33-year period at Site 801, while at Site 804 it increased by 2.0 °C, and at all other sites it increased by an average of 0.8 °C. Most of the increase in water temperature occurred during the drought of 1997–2009.





**Figure 2.** Space-time plots of environmental variables: (a) electrical conductivity (EC), (b) turbidity, (c) water temperature, (d) pH, (e) flow, (f) rainfall, and (g) air temperature. Lines represent the deseasonalized component of the fitted generalized additive models, to highlight the longer-term patterns. Color indicates site distance from source (km). Sites 808, 811, and 812 at distances 1,389, 1,737, and 1,910 km are located within the semiarid region. Periods of drought and flood are indicated with horizontal lines beneath the time axis.



**Figure 3.** Space-time plots of (a) total abundance and (b) taxa richness. Lines fitted by generalized additive models. Color indicates site distance from source (km). Sites 808, 811, and 812 at distances 1389, 1737, and 1910 km are located within the semiarid region. Periods of drought and flood are indicated with horizontal lines beneath the time axis.

### 3.2. Abundance and Richness of the Macroinvertebrate Community

Taxa richness and total abundance of macroinvertebrates began to increase across all sites after the 1993 flood and declined after peaking during the Millennium drought (Figure 3). Abundance displayed a unimodal response to EC with a peak at about 300  $\mu$ S/cm, whereas it decreased with increasing turbidity, varied seasonally with temperature, and increased with time after the 1993 flood (Figures 4a–4d). Richness decreased with increasing EC and pH, whereas it varied seasonally with temperature, and it increased with time after the 1993 flood (Figures 4e–4h).

### 3.3. Macroinvertebrate Community Structure

Various diagnostic tools were used to determine the number of PCO axes that potentially contain interpretable structure. Scree plots showing the results obtained by each of these methods are given in Figure 5. The first three methods compare the observed distribution of eigenvalues to a distribution expected for unstructured multivariate data having the same number of dimensions, and the fourth uses bootstrapped confidence intervals of the eigenvalues. The broken-stick model (Figure 5a) and conditional random permutation method (Figure 5c) suggest that there were 11 or 12 nontrivial axes out of a total of 268. The holistic random permutation method (Figure 5b) suggests that there were five or six nontrivial axes, and the bootstrapped eigenvalues (Figure 5d) indicate that there were two nontrivial axes. Space-time plots of individual PCOs indicated there was spatial and temporal structure in the first four PCOs, including a



**Figure 4.** Plots of smooth components of GAMs for total abundance (a-d) and taxa richness (e-h) against the corresponding explanatory variable. The effective degrees of freedom for each smooth term is given in parentheses on the *y* axis (higher values indicating greater *wiggliness* of the smooth), and the rug plots (small tick marks above the *x* axes) indicate values of the explanatory variable. EC = electrical conductivity.

seasonal pattern in the third and fourth axes, but there was no clear spatial structure in subsequent axes, only a gradually decreasing seasonal pattern up to PCO12.

On the basis of patterns in the first four PCOs (Figure 6), a spatiotemporal dbRDA model was fitted to the data. Orthogonal polynomial terms in the model accounted for the quadratic relationship between community structure and distance from source (particularly apparent in PCO1; see Figure 6a), a cubic polynomial trend over time, and the interaction between the two. A cosine term was added to account for seasonality. All parameter estimates were statistically significant at the 0.05 level. However, there was some evidence from residual diagnostic plots that Site 804 did not entirely conform with the spatial component of the model, which we attribute to an effect of the Yarrawonga weir (4 km upstream) on the basis of an earlier study (Bennison



**Figure 5.** Scree plots for the principal coordinate (PCO) analysis with (a) the broken-stick model, (b) 95% permutation confidence intervals from a null model of no structure, where percentages represented by each axis are taken holistically as a fraction of the total variation; (c) as in (b), except percentages represented by each axis are taken conditionally as a fraction of the variation remaining after removing that which is explained by prior axes; and (d) 95% bootstrap confidence intervals for eigenvalues.

et al., 1989). The indicator variable [dist = 527] was therefore added to the model to account for the shift in community structure at this site. The analysis of variance table for the final model is given as Table 1. The dbRDA model explained 24% of the total variation. Fitted values from the model are overlayed on the space-time plots for the first four PCOs (Figures 6a–6d), with Spearman correlations between each PCO axis and the environmental variables provided above each plot.

PCO1 represented 9.1% of the total variation in the macroinvertebrate community. The primary pattern in PCO1 is a quadratic spatial trend along the Murray River, and a convergence of PCO sample scores toward those of the mid-Murray sites over time (Figure 6a). This pattern appears consistent with that of rainfall and water temperature, and PCO1 was positively correlated with temperature (weakly) and negatively with rainfall. PCO1 was also moderately positively correlated with pH, turbidity, and EC.

PCO2 represented 7.3% of the variation in the macroinvertebrate community and displayed a spatial trend along the river (Figure 6b). While Site 814 remained in the higher region of the PCO2 scores, there again was a pattern of increasing similarity of PCO2 scores among sites over time, indicating a loss of spatial variability of community structure throughout the river system. EC was most highly correlated with PCO<sub>2</sub>, followed by turbidity and pH. This suggests that PCO2 reflects changes in community structure attributable to water quality.

PCO3 and PCO4 represented 4.6% and 3.7% of the variation, respectively (Figures 6c and 6d). Unlike later axes, these two axes still contain an appreciable spatial component. Spatially, PCO3 appears to reflect a difference between the lowermost site and all other sites, whereas PCO4 appears to distinguish the two



**Figure 6.** Space-time plots for principal coordinate (PCO) axes 1–4 (a–d, respectively) with fitted values (lines) from the distance-based redundancy analysis model with space and time as predictors. Spearman rank correlations between each PCO axis and the environmental variables are given above each graph. Color indicates site distance from source (km). Sites 808, 811, and 812 at distances 1,389, 1,737, and 1,910 km are located within the semiarid region. Periods of drought and flood are indicated with horizontal lines beneath the time axis. EC = electrical conductivity.

uppermost sites from lower sites. Both axes exhibit cubic polynomial trends in time, which were also evident in the first two axes and might be related in part to the 1993 flood, and a seasonal pattern that was not evident in the first two PCO axes.



Source	Df	SS	F	р
Poly (Dist, 2)	2	11.6	16.4	0.001
Cos ( $\pi \times$ Time)	1	1.8	5.0	0.001
Poly (Time, 3)	3	5.5	5.2	0.001
[Dist = 527]	1	1.8	5.2	0.001
Poly (Dist, 2) $\times$ Poly (Time, 3)	6	5.2	2.4	0.001
Poly (Time, 3) $\times$ [Dist = 527]	3	1.9	1.8	0.001
Residual	252	88.9		

Note. ANOVA = analysis of variance.

The macroinvertebrate community composition was then modeled with dbRDA as a function of EC, turbidity, water temperature, pH, TKN, TP, normalized flow, rainfall, temperature, and both distance and time. Scatter plots of PCOs against environmental variables indicated the relationships were approximately linear, and distance and time were included in the dbRDA model in the same forms as in the previous spatiotemporal model. Some variables or terms were dropped from the model on the basis of significance tests, multicollinearity diagnostics, and the AIC/BIC (see Akaike and Bayesian Information Criteria for distance-based linear models in Anderson et al., 2008); the final model included EC, turbidity, water temperature, normalized flow, pH, rainfall, the indicator variable for Site 804, and a cubic term for time. The analysis of variance table for

the final model is given as Table 2. The model explained 16% of the total variation, with the main sources of variation being EC and the polynomial trend in time. A conditional dbRDA indicated that the environmental variables (not including the indicator variable or time) accounted for 10% of the variation in community structure, while the terms from the previous spatiotemporal model explained a further 16% of the variation. This indicates that the environmental variables, for which data were available, account for only 38% of the spatiotemporal variation and thus provide only a partial explanation of the observed spatiotemporal patterns.

### 3.4. Taxa Response Curves Along Ordination Axes

Shade plots displaying the response curves of individual taxa along the first four PCOs are given as Figure 7. The taxa are clustered in 30 coherent groups (A–AD), according to the similarity in their patterns of abundance through space and time. It is evident from Figure 7 that some taxa have a narrow coverage of the underlying environmental gradients whereas others span the full range of those gradients and that the shapes of the curves are generally either monotonic or unimodal. The ordering of the coherent groups accords principally with the first two PCOs, which represent variation in the macroinvertebrate assemblage that we attribute primarily to responses to climate (rainfall and temperature) and water quality (EC and turbidity). The explained deviances for individual smooth PCOs in the GAMs (Table 3) similarly indicate that coherent group abundances load mostly on the first two PCOs. Figure 7 also indicates there are ecological important patterns in PCO3 and PCO4, in groups L, K, B, P, O, and C, in particular. Group P, in fact, loads almost exclusively on PCO3, as shown in Table 3. Nine coherent groups that were most strongly associated with the patterns in the ordination axes (Table 3), and which span the ordering shown in Figure 7, were chosen for a more detailed examination.

From Figure 7, taxa from groups K and L are generally restricted to sites with higher rainfall, lower water temperature, and good water quality. These groups occurred mainly in the uppermost site. Groups K and L were composed of taxa that are generally associated with cool, well oxygenated, flowing water, and typically considered pollution sensitive, such as *Coloburiscoides* sp. and *Atalophlebia* spp. (mayflies), *Leptoperla* spp. (stone-

#### Table 2

ANOVA Table for dbRDA Mod	lel With Environmental	Variables, and	Space and
Time as Predictors			

Df	SS	F	p
1	5.1	13.4	0.001
1	1.2	3.3	0.001
1	1.9	5.0	0.001
1	1.2	3.1	0.001
1	1.8	4.7	0.001
1	0.9	2.5	0.001
1	1.6	4.1	0.001
3	5.2	4.6	0.001
258	97.8		
	Df 1 1 1 1 1 1 3 258	Df SS   1 5.1   1 1.2   1 1.9   1 1.2   1 1.2   1 1.2   1 1.2   1 1.6   3 5.2   258 97.8	Df SS F   1 5.1 13.4   1 1.2 3.3   1 1.9 5.0   1 1.2 3.1   1 1.2 3.1   1 1.8 4.7   1 0.9 2.5   1 1.6 4.1   3 5.2 4.6   258 97.8 97.8

*Note*. ANOVA = analysis of variance; dbRDA = distance-based redundancy analysis.

flies), *Cheumatopsyche* spp. (net-spinning Caddisfly), *Austrosimulium* spp. (filter-feeding diptera), and *Simsonia* spp., *Notriolus* spp. and *Austrolimnius* spp. (riffle beetles). These two groups generally increased in abundance at Site 801 from the 1990s and appeared to decrease near the end of the study period (e.g., see Figure 8a).

Groups E, C, and P displayed a general increase in abundance, more so at one particular site (different for each group), from around 1990 onward. Taxa within group E cover a range of climates but generally occur where the water quality is better. This group was most abundant at Sites 801 and 804. Group E comprised three, highly mobile, predatory taxa, including one dragonfly species (*Apocordulia macrops*) that is considered pollution sensitive, with a SIGNAL2 score of 10 (SIGNAL2 scores ranging from 1 for pollution tolerant to 10 for pollution sensitive have been ascribed to most families of Australian macroinvertebrates; see Chessman, 2003), as well as a nonbiting midge (*Rheotanytarsus* spp.) that is considered



Figure 7. Shade plots of fourth-root abundances against principal coordinates (PCOs) 1–4. Coherent species groups can be identified from the colored symbols shown in the legend at top left. A common abundance scale, based on fourth-root abundances, was used for all plots.

Table 3
GAMs of Coherent Group Fourth-Root Average Abundances Against PCOs

	Explained deviance per smooth term					
Group	s (PCO1)	s (PCO2)	s (PCO3)	s (PCO4)	Explained deviance (%)	Null deviance
L	41.7	13	1.8	6	89.8	402.6
Р	6.7	2	35.6	3.7	88.4	1146.8
К	23.4	10.5	1.2	3	87.1	333.8
AA	10.8	14.6	0.3	2.1	84.7	1720.8
AD	9	15	8.1	5.7	76.2	1016.4
Т	20.4	1.4	1.5	1.6	76	1023.7
G	21.3	3	8.5	0.2	75.3	339.5
E	16	16	1.1	6.4	72.5	357.6
AC	11.3	7.1	6	3.6	66.5	720.5
С	10.6	7.1	7.8	6.8	65.7	314.4
0	1.9	10.9	2.5	9.3	64.7	355
U	9.1	0	10.3	-0.1	61.1	615.2
F	3.7	3.8	1.9	3.4	59.8	87.3
1	18.7	4.1	6.7	1.3	56.7	63.9
J	0.2	1.3	4.6	4.2	55.8	137.1
Q	9.5	0.3	3.5	0.5	55.7	272.4
Α	1.2	3.5	8.1	3.4	53.4	73.2
М	4.4	1.1	0.9	3.6	53.2	96.2
V	2.6	4.5	4.2	1.2	53.1	507.8
D	10.9	0.6	11.4	10.3	52.9	198.3
Н	1.1	7.4	10.7	2.9	47.3	96.6
В	3.7	12.4	0	4	47.1	310.8
Y	2.6	4	1.5	0.4	44.2	390
Z	4.5	3.2	0.9	0.9	43.2	546
R	1.4	2.2	2.4	2.7	42.8	332.1
Ν	2.8	12.5	0.6	10.5	39.2	100.3
S	1.2	2.6	1.4	1.3	32.7	391.5
Х	3.7	0.6	0.9	0.2	26.7	340.6
W	1.2	0.1	0.6	0.3	21.3	299
AB	0.1	3.8	2.8	0.7	19.1	218.8

*Note.* GAM = generalized additive model; PCO = principal coordinate.

tolerant to pollution (SIGNAL2 score of 3). Group C occurred at warmer climates and all but the poorest of water quality. Group C, which included the freshwater yabby (*Cherax destructor*), a chironomid (*Kiefferulus* spp.), and a caddisfly (*Hydroptila* spp.), occurred mostly at the midreach site 811 (1,737 km) and gradually increased in abundance from 1990. Group P covered a range of climates and water quality, except for *Amarinus lacustris* and Hydrobiidae which were restricted to the higher salinities. Group P (Figure 8b) increased in abundance across all sites, with a rapid population increase in the early 1990s at Site 814, followed by a decrease again in the mid-2000s to become comparable in abundance to the other sites. This group consisted of highly tolerant, benthic taxa, including two snails, a flat worm, a proboscis worm, and a freshwater crab. The freshwater crab (*Amarinus lacustris*) occurred exclusively at Site 814, and the mud snail (Hydrobiidae) occurred almost exclusively at this site. These two taxa are known to inhabit brackish waters.

Groups AA, AC, and AD exhibited a wide coverage of climate and water quality, but with a slight bias toward warmer temperatures and a tendency to avoid the more saline and turbid waters. These groups exhibited an increase in abundance over time (Figure 8c), and all displayed the quadratic spatial pattern seen in PCO1, with higher abundances in the mid-Murray. However, the magnitude of this spatial pattern diminished over time for Group AA as abundances increased mainly at the upper and lower sites, resulting in a convergence with the mid-Murray sites. Groups AA, AC, and AD are composed of highly pollution-tolerant taxa with low SIGNAL2 scores (2–4). Chironomidae made up the majority of taxa, but these groups also contained Oligochaeta (worms), Nematoda (roundworms), and Caenidae (mayflies), which were the most numerous taxa collected overall.

Many of the taxa in these groups increased in their frequency of occurrence after the flood of 1993. These included taxa from Ephemeroptera (Caenidae), Plecoptera (Gripopterygidae), Trichoptera (Hydroptilidae), Odonata (Protoneuridae, Austrocorduliidae), Diptera (Chironimidae, including the subfamily Tanypodinae and tribes Chironomini and

Tanytarsini, Orthocladiinae, and Simuliidae), Nematoda, Crustacea (Parastacidae), Mollusca (Planorbidae, Hydrobiidae, and Ancylidae), Acarina (Halacaridae), Hydrozoa (Hydridae), Coleoptera (Elmidae and Dytiscidae), and Tricladida (Dugesiidae). Apart from a few predatory species within Tanypodinae, Dytiscidae, Odonata, and Dugesiidae, most of which occurred within group E at Site 804, representatives from the other families were mostly shredders, collectors, and scrapers that feed on detritus and plants, and the majority have more than one generation per year (MDFRC, 2013; Schäfer et al., 2011).

# 3.5. Counterfactual Analysis to Elucidate how Environmental Changes Have Influenced Groups of Taxa

The predicted abundances of the 30 coherent groups of taxa under various scenarios are shown in Figure 9 for a selection of sites in winter 2011. The purpose of this analysis was to assist in understanding what the community structure might have been near the end of the study period if certain conditions were different, including EC. Winter 2011 was chosen for this analysis because it was the only sampling occasion near the end of the study period that had macroinvertebrate data for all six sites; though, it was repeated for summer 2010 with a subset of sites, and similar patterns were observed. In the following we focus mainly on the nine coherent groups described in the previous section.

The scenarios for Site 801 included predictions under actual conditions (i.e., the observed abundances), a pH that was 0.3 units higher, and a water temperature that was lower by 2.5 °C. The actual conditions in 2011 are given in Table 4. Results suggest that groups E, AC, and AD would have been in higher





**Figure 8.** Space-time plots of averaged fourth-root abundances for selected coherent groups of taxa: (a) Group L, (b) Group P, and (c) Group AA. Color indicates site distance from source (km). Sites 808, 811, and 812 at distances 1,389, 1,737, and 1,910 km are located within the semiarid region. Periods of drought and flood are indicated with horizontal lines beneath the time axis.

abundance if the pH was 0.3 units higher, that is, if the pH had been about 7.3 as it was in 1980. Had the water temperature been 2.5 °C cooler in 2011 abundances would have been higher for groups K and L, and lower for groups E, AC, and AD.

The shift in community structure at site 804, which we attributed to the proximity of Yarrawonga weir 4 km upstream, appears to be due to a lower abundance of several groups of taxa, including groups AA and AC, whereas groups C and E were in higher abundance. This was inferred by setting the indicator variable [dist = 527] to zero in the dbRDA model then predicting abundances from the new PCO scores resulting from the change. For simplicity we refer to this intervention as *no weir* in Figure 9. Group E would have been more abundant if the water temperature had been 2.0 °C cooler, whereas Groups C, AC, and AD would have been less abundant.

The most notable scenario at Site 814 is that several groups of taxa would have been more or less abundant if the EC had have been higher by 150  $\mu$ S/cm, as it was predicted to be at Morgan (between Sites 812 and 814) in the absence of salinity mitigation strategies, or 400  $\mu$ S/cm higher, as it was in 1980. Group P would have been more abundant, whereas groups E, AA, AC, and AD less abundant. Furthermore, the GAMs presented in Figures 4a and 4e suggest that total abundance and taxa richness would have been lower under these two scenarios of higher EC, although 95% confidence intervals for predictions suggest that the evidence is weak. Predictions with approximate 95% confidence intervals (in parentheses) under actual conditions, and EC higher by 150 and 400  $\mu$ S/cm are, for abundance, 1,366 (750, 1982) individuals, 999 (527, 1471) individuals, and 628 (284, 972) individuals, respectively; and for richness, 33 (30, 36) taxa, 32 (29, 35) taxa, and 31 (27, 34) taxa. Of these, the only difference in means that is statistically significant is between the mean



**Figure 9.** Plots of averaged fourth-root abundances for coherent groups of taxa, at selected sites –(a) Site 801, (b) Site 804, and (c) Site 814 –under various scenarios: (i) observed abundances under the actual conditions of the winter of 2011 (see Table 4); (ii) predicted abundances if the pH had been 0.3 units higher at Site 801; (iii) predicted abundances if the water temperature had been cooler by 2.5, 2.0, and 0.8 °C for Sites 801, 804, and 814, respectively; (iv) predicted abundances if the indicator variable for Site 804 was set to zero, which for simplicity we refer to here as no weir (at Yarrawonga); and (v) predicted abundances if the electrical conductivity had been higher by 150 and 400  $\mu$ S/cm at Site 814.

Table 4	
Actual Conditions in Winter 2011	

Variable	Site 801	Site 804	Site 814
EC (µS/cm)	44	71	373
Turbidity (NTU)	4	14	126
Water temperature (°C)	11.7	15.5	15.1
Normalized flow	0.02	0.13	1.54
рН	7.0	7.9	7.5
Rainfall (mm/day)	1.4	0.7	1.5

*Note*. EC = electrical conductivity.

abundance under actual conditions and the mean abundance at 400  $\mu$ S/cm higher EC, with the approximate 95% confidence interval for the difference being (46, 1428) individuals.

### 4. Discussion

The WQMP began in 1980 with the purpose of evaluating measures to control salinity and nutrients and restore flows, as well as detecting other problems that might be affecting aquatic ecosystems of the Murray River. Since its inception there have been several major flooding events, with widespread flooding in 1993 and 2010–11, and droughts,

including the 1982–1983 drought and the Millennium drought from 1997 to 2009. Taxa richness and total abundance of macroinvertebrates increased across all sites after the 1993 flood then began to decline after peaking during the Millennium drought. Richness decreased with increasing EC and pH, whereas it varied seasonally with temperature, and it increased with time after the mid-1990s. Abundance displayed a unimodal response to EC with a peak at about 300 µS/cm, whereas it decreased with increasing turbidity, varied seasonally with temperature, and increased with time after the mid-1990s. The changes in community structure were similarly related to EC, pH, turbidity, and water temperature and also to normalized flow and rainfall. The temporal trends in the univariate and multivariate analyses indicate something other than the measured environmental variables was responsible for the persistent changes in richness, abundance, and community structure following the flood of 1993. Moreover, in the multivariate analysis, the environmental variables (EC, turbidity, water temperature, normalized flow, and rainfall) accounted for only 38% of the systematic variation through space and time. This suggests that flood was a major determinant of community structure, which accords with the stream literature (Death, 2008; Lake, 2000; Reice et al., 1990; Resh et al., 1988).

These observations are consistent with our first hypothesis: that the number of taxa would increase in response to floods, in agreement with the intermediate disturbance hypothesis. However, we also observed an increase in total abundance. It is well known that floods can have many indirect effects on the structure of macroinvertebrate communities and that these can manifest over a range of time scales. Numerous studies have reported that richness and density decrease in response to high flows—as individuals are washed out, crushed by moving substrate and debris, or take refuge—and that recovery generally happens within a few weeks (Brooks & Boulton, 1991; Death, 2008; Mannes et al., 2008; Robinson, 2012; Scrimgeour & Winterbourn, 1989). Floods can also have a major effect on habitat—including scouring and aggradation/degradation of streambeds, removal of plants, nutrient input, and deposi-Ozaki, 1996). Natural floods in the Rhône River resulted in an increase in richness (Daufresne et al., 2007; Fruget et al., 2001) and abundance of some macroinvertebrate taxa (Daufresne et al., 2007), particularly the abundance of eurytolerant and invasive taxa, that persisted for more than a decade without any apparent sign of recovery. We found that the immediate effect of high flows was negligible, which might be a result of the long deployment periods for artificial substrates, but that the 1993 flood appeared to have a sizeable longer-term impact. Daufresne et al. (2007) hypothesized a switch to an alternative stable state (Scheffer et al., 2001) as an explanation for the persistent increase in richness observed in their study, but in our study there were signs of recovery after about a decade, and we suspect the explanation lies in the indirect effects of floods on habitat (e.g., see Townsend & Scarsbrook, 1997).

Apart from a few predators, all of the taxa that became more prevalent after the 1993 flood were shredders, collectors, and scrapers that feed on detritus and plants, and the majority have more than one generation per year (MDFRC, 2013; Schäfer et al., 2011). This could indicate that the flood had a long-lasting effect on community structure through the promotion of taxa capable of exploiting inputs of detritus and debris from the floodplain (Death, 2008; Junk et al., 1989), and possibly the periphytic algal biomass that might have increased in response to nutrient inputs (Baldwin & Mitchell, 2000; Cummins & Klug, 1979; Rier & Stevenson, 2006). In the longer term, future monitoring could include variables thought to mediate the effects of floods on community structure—such as substrate composition, periphytic algal biomass, and

both fine and coarse particulate organic matter—but, in the short term, the models presented here could be further developed to include lagged flows that might account for the delayed effects of floods.

In addition to the effects of floods, we hypothesized that warming would alter the geographical ranges of some species and shift the community toward eurytolerant taxa. Water temperature increased by about 2.5 °C over the 33-year period at Site 801, while at Site 804 it increased by 2.0 °C, and at all other sites it increased by an average of 0.8 °C. There was evidence that the increase in water temperature of 2.5 °C at Site 801 reduced the abundance of Groups K and L, including taxa such as Coloburiscoides sp. and Atalophlebia spp. (Ephemeroptera), Leoptoperla spp. (Plectoptera), and Cheumatopsyche spp. (Trichoptera), all of which are generally associated with cool water and occur in relatively high numbers at Site 801 in winter. Conversely, Groups AC and AD-which include tolerant taxa from Chironomidae, Oligochaeta, Nematoda, and Caenidae that are generally found across a range of sites—were in higher abundances. For example, Nematoda and Cladopelma curtivalva (Chironomidae) were rarely detected at the upper sites until the onset of the Millennium drought in the summer of 1996 and are now present in quite high numbers. Similarly, Polypedilum spp. (Chironomidae) were in very low numbers until the onset of the drought. Prior to the drought, these taxa were mostly abundant in the warmer midreaches of the river. This would suggest a shift in the geographic range of these taxa, although abundances generally decreased again when the drought ended. These findings are in agreement with the majority of long-term studies of macroinvertebrates (Chessman, 2009; Daufresne et al., 2007; Durance & Ormerod, 2007; Floury et al., 2013; Fruget et al., 2001, 2015).

The increase in water temperature in the Murray River appears to have occurred during the drought of 1997–2009, after which it began to fall again. There was no clear directional trend in the air temperature data through time, only an apparent long-term cycle with a trough occurring in the early 1990s, and this pattern was consistent across all sites. Further, the observed increases in water temperature are consistent with those reported by others during times of drought (see Mosley, 2015), which have ranged from 1.3 to 7 °C. Accordingly, we believe that the variation along the river in water temperature increases is largely due to local factors such as lower water levels and reduced groundwater inputs.

Our third hypothesis concerned the response of the macroinvertebrate community to decreasing salinity levels. EC decreased in the Lower Murray over the 33-year period—at Site 814 it decreased from an average of 800 to about 400 µS/cm—and this can, in part, be attributed to salinity mitigation strategies. Species richness is believed to peak at an EC of between 300 and 490 µS/cm, with a statistically significant decline at <99 and >1,500  $\mu$ S/cm (Kefford et al., 2011). Kefford et al. (2011) found that family richness similarly appears to peak at 300–490  $\mu$ S/cm, but with a statistically significant decline detected only at >4,000  $\mu$ S/cm. Our results indicate that richness (mostly at the level of genus and family) decreases with increasing EC, but abundance displayed a unimodal response with a peak at about 300 µS/cm. Moreover, counterfactual predictions indicate that many taxa had increased in abundance as a result of reduced salinities, while some decreased. For example, Cherax destructor (Crustacea) and Hydroptila spp. (Trichoptera) began to appear with increasing abundance in the Lower Murray from the early 1990s. Hydroptila spp. are known to have a relatively low tolerance to salinity (Horrigan et al., 2005; Rutherford & Kefford, 2005), whereas Cherax destructor survives experimentally in salinities up to 25 g/L (Mills & Geddes, 1980) and has an estimated maximum field salinity tolerance of 16.1 g/L (Rutherford & Kefford, 2005), though anecdotal evidence suggests it tends to favor salinities between 100 and 1,580 mg/L (approx.780-2,500 µS/cm) in the field (Hart et al., 1991). Conversely, the freshwater crab Amarinus lacustris, which is known to inhabit freshwaters with higher salinities and only occurred at Site 814, appears to have decreased in abundance with the lower salinity. This suggests that the geographic ranges of some taxa may be shifting in response to changing salinity in the Lower Murray, due to either a direct sublethal effect or an indirect effect mediated by biotic interaction.

Unexpectedly, we also observed an appreciable decrease in pH at the uppermost site (Site 801). The pH at this site ranged between 6.3 and 7.8, with the mean varying from about 7.3 in the 1980s to 6.6 in 2005, and increasing again to 7.0 in the 2010s. The pH levels below 5.0 and above 9.0 are known to be toxic to most macroinvertebrates (Hussain & Pandit, 2012), but within this range Graca et al. (2004) found that richness decreases with increasing pH. We similarly observed a decrease in richness with increasing pH. We also found that some taxa at Site 801 were in lower abundance at the lower pH levels, including from

the Chironomidae tribes Tanytarsini and Chironomini, and subfamilies Orthocladiinae and Tanypodinae, and also Coenagrionidae (Odonata), and Hydroptilidae and Economidae (Trichoptera). It may be that these taxa have pH optima above 7 (Tripole et al., 2008). Site 801 lies within a subcatchment known to suffer from soil acidification (Chapman et al., 2007), and the lower pH in the Murray River might be the result of acid runoff during periods of high rainfall (Billett & Cresser, 1992; Krug & Frink, 1983). The use of lime to counter soil acidification has increased exponentially in this region since the early 1990s (Scott et al., 2007), and this, together with lower runoff during the Millennium drought, might explain the increasing pH at Site 801 from the mid-2000s.

Floods, droughts, climate, land use, weirs, and water quality, therefore, appear to have contributed to the spatial and temporal changes in macroinvertebrate communities. The taxonomic data presented here indicate two dominant spatial patterns: one being a quadratic trend that is, at least partly, related to climate (rainfall and temperature); and the other a monotonic trend related to water quality (particularly EC). There was also a noticeable decrease below the Yarrawonga weir in the abundance of gathering collectors (Cummins & Klug, 1979), which feed on detritus in sediments, and this accords with the serial discontinuity concept (Ward & Stanford, 1983). Over time, the spatial trends diminished, as taxa from the mid-Murray expanded their range and communities converged due to increases in temperature at the uppermost sites and decreases in salinity at the lower sites. It is possible that the 1993 flood also contributed to the convergence of communities over time, especially taxa within group AA such as dipterans, mayflies, and perhaps nematodes. Importantly, against this backdrop of climate and water quality variability, we believe we have been able to distinguish a continuum of change in community structure along a salinity gradient, and that salinity mitigation has contributed to the changes that have occurred over the 33-year period of this study.

Of the management interventions that have been implemented since the 1980s—salinity mitigation, reducing nutrient pollution, capping water diversions and allocating water for environmental flows (see section 1)—we can be more certain about the impact of salinity mitigation principally because an estimate of the reduction in salinity that resulted was available from a previous study. Our results suggest that rising salinity had impacted macroinvertebrate communities in the past and that mitigation strategies may be effective in reversing that impact. We acknowledge there is some uncertainty in the modeled improvement in EC attributed to salinity strategies (MDBMC, 2015), but the reported contribution of 150 µS/cm is an appreciable proportion of the average reduction of 400 µS/cm in the Lower Murray over the study period. We also acknowledge that there were spatial and temporal patterns in the communities that we were not able to explain with the available data, and that confounding bias might therefore be an issue. Further research is needed to identify the processes underlying these patterns if we are to reduce uncertainty in the estimated effects of salinity and mitigation strategies, as well as detect other anthropogenic impacts. This is an important next step because salt loads are projected to increase in the MDB (MDBMC, 2015), and elsewhere around the world (Cañedo-Argüelles et al., 2013), and a better understanding of the ecological risks is needed to guide future policy and programs. It is less clear from our results whether pollution abatement and environmental watering strategies have had an impact on macroinvertebrate communities. While there appeared to be a gradual reduction in TKN and TP over the 33 years, it is not possible to attribute this to the algal (and nutrient) management strategy owing to the difficulty in quantifying the effect of this intervention (MDBMC, 1994). Further, both TKN and TP were highly correlated with turbidity, making it difficult to separate their effects on macroinvertebrates in the modeling that was done, and unfortunately, measurements of soluble nutrient concentrations (nitrate, nitrite, and filterable reactive phosphorus) were infrequent in the data set that was available. With regard to environmental flows, we know that regulation of the Murray River has reduced the magnitude of average annual floods (annual exceedance probability 50%) by more than 50%, but big floods (average recurrence interval of 20 years or more) have been little affected (Maheshwari et al., 1995). Given the apparent importance of large floods on macroinvertebrate communities, evident from our analysis, it is unclear whether environmental flows have had or could have an appreciable effect on in-channel macroinvertebrate communities, especially with the constraints placed on such flows to avoid inundation of private property and infrastructure (MDBA, 2014). Further research is required on the (delayed) effects of floods of various magnitudes to better understand the impact of floods and environmental flows on the aquatic macroinvertebrates of the Murray River.



#### Acknowledgments

We wish to acknowledge the many people who have contributed to this body of work, including those who have contributed to the collection and processing of samples over the 33-year period from the Murray Darling Freshwater Research Centre, Water ECOscience, and the Australian Water Quality Centre; staff of SA Water and the MDBA who provided the water quality and hydrological data; and staff of the MDBA whose strong support for this project has enabled its continuation, including Norm Mackay, Brian Lawrence, Martin Shafron, Mark Vanner, Richard Moxham, Kris Kleeman, Rob Kingham and Tapas Biswas, Garv Bennison, Terry Hillman, and the late Peter Waller who were instrumental in establishing the Murray Monitoring Program. This work was funded by the Murray Darling Basin Authority (MDBA: formerly the Murray Darling Basin Commission and River Murray Commission). The data used in this publication are available online at Figshare (Paul et al., 2018).

### References

- Anderson, M. J., Gorley, R. N., & Clarke, K. R. (2008). PERMANOVA+ for PRIMER: Guide to software and statistical methods. Plymouth, UK: PRIMER-E Ltd.
- Anderson, M. J., & Legendre, P. (1999). An empirical comparison of permutation methods for tests of partial regression coefficients in a linear model. Journal of Statistical Computing and Simulation, 62(3), 271–303. https://doi.org/10.1080/00949659908811936
- Angradi, T. R., Bolgrien, D. W., Jicha, T. M., Pearson, M. S., Hill, B. H., Taylor, D. L., et al. (2009). A bioassessment approach for mid-continent great rivers: The Upper Mississippi, Missouri, and Ohio (USA). *Environmental Monitoring and Assessment*, 152(1-4), 425–442. https://doi.org/ 10.1007/s10661-008-0327-1
- Australian Bureau of Statistics (2008). 4610.0.55.007—Water and the Murray-Darling Basin—A statistical profile, 2000–01 to 2005–06. Retrieved from http://www.abs.gov.au%2FAUSSTATS%2Fabs%40.nsf%2FLookup%2F4610.0.55.007Main%2BFeatures12000%26ndash% 3B01%2520to%25202005%26ndash%3B06%3FOpenDocument
- Baldwin, D. S., & Mitchell, A. M. (2000). The effects of drying and re-flooding on the sediment and soil nutrient dynamics of lowland riverfloodplain systems: A synthesis. *Regulated Rivers: Research & Management, 16*(5), 457–467. https://doi.org/10.1002/1099-1646(200009/ 10)16:5<457::AID-RRR597>3.0.CO;2-B
- Bennison, G. L., Hillman, T. J., & Suter, P. J. (1989). Macroinvertebrates of the River Murray (survey and monitoring: 1980–1985). Canberra, Australia: Murray-Darling Basin Commission.
- Billett, M. F., & Cresser, M. S. (1992). Predicting stream-water quality using catchment and soil chemical characteristics. Environmental Pollution, 77(2-3), 263–268. https://doi.org/10.1016/0269-7491(92)90085-O
- Blackmore, D. J., Goss, K. F., Newman, R. J., Powell, J. R., & Sharma, P. (2000). Salinity management in the Murray-Darling Basin. Paper presented at the 10th World Water Congress: Water, the Worlds Most Important Resource, Melbourne, Australia. Retrieved from https://search. informit.com.au/documentSummary;dn=517246577551518;res=IELENG
- Bradley, D. C., & Ormerod, S. J. (2001). Community persistence among stream invertebrates tracks the North Atlantic Oscillation. *Journal of Animal Ecology*, 70(6), 987–996. https://doi.org/10.1046/j.0021-8790.2001.00551.x
- Brooks, S. S., & Boulton, A. J. (1991). Recolonization dynamics of benthic macroinvertebrates after artificial and natural disturbances in an Australian temporary stream. *Australian Journal of Marine and Freshwater Research*, *42*(3), 295–308. https://doi.org/10.1071/MF9910295 Bureau of Meteorology (2012). Climate data online. Retrieved from http://www.bom.gov.au/climate/data/
- Cañedo-Argüelles, M., Kefford, B. J., Piscart, C., Prat, N., Schäfer, R. B., & Schulz, C.-J. (2013). Salinisation of rivers: An urgent ecological issue. Environmental Pollution, 173, 157–167. https://doi.org/10.1016/j.envpol.2012.10.011
- Chapman, G., Davy, M., Symes, L., Yang, X., & Wilson, B. (2007). Murray catchment soil acidification monitoring trial. Canberra: National Land & Water Resources Audit.
- Chessman, B. C. (1995). Rapid assessment of rivers using macroinvertebrates: A procedure based on habitat-specific sampling, familylevel identification and a biotic index. *Australian Journal of Ecology*, 20(1), 122–129. https://doi.org/10.1111/j.1442-9993.1995. tb00526.x
- Chessman, B. C. (2003). New sensitivity grades for Australian river macroinvertebrates. Journal of Marine and Freshwater Research, 54(2), 95–103. https://doi.org/10.1071/MF02114
- Chessman, B. C. (2009). Climatic changes and 13-year trends in stream macroinvertebrate assemblages in New South Wales, Australia. *Global Change Biology*, *15*, 1–12.
- Clarke, K. R., & Gorley, R. N. (2015). PRIMER v7: User manual/tutorial. Plymouth, UK: PRIMER-E.
- Clarke, K. R., Gorley, R. N., Somerfield, P. J., & Warwick, R. N. (2014). Change in marine communities: An approach to statistical analysis and interpretation, (3rd ed.). Plymouth, United Kingdom: PRIMER-E.
- Clarke, K. R., Tweedley, J. R., & Valesini, F. J. (2014). Simple shade plots aid better long-term choices of data pre-treatment in multivariate assemblage studies. *Journal of the Marine Biological Association of the UK*, 94(01), 1–16. https://doi.org/10.1017/ S0025315413001227
- Connell, J. H. (1978). Diversity in tropical rain forests and coral reefs. *Science*, 199(4335), 1302–1310. https://doi.org/10.1126/ science.199.4335.1302
- Crase, L. (2008). An introduction to Australian water policy. In L. Crase (Ed.), Water policy in Australia: The impact of change and uncertainty (Chapter 1, pp. 1–16). Washington, DC: RFF Press.
- CSIRO (2012). Climate and water availability in south-eastern Australia: A synthesis of findings from Phase 2 of the South Eastern Australian Climate Initiative (SEACI). Australia: CSIRO. Retrieved from http://www.seaci.org/publications/documents/SEACI-2Reports/SEACI\_Phase2\_ SynthesisReport.pdf
- Cummins, K. W., & Klug, M. J. (1979). Feeding ecology of stream invertebrates. Annual Review of Ecology and Systematics, 10(1), 147–172. https://doi.org/10.1146/annurev.es.10.110179.001051
- Cunningham, R. B., & Morton, R. (1983). A statistical method for the estimation of trend in salinity in the River Murray. Australian Journal of Soil Research, 21(2), 123–132. https://doi.org/10.1071/SR9830123
- Daufresne, M., Bady, P., & Fruget, J. (2007). Impacts of global changes and extreme hydroclimatic events on macroinvertebrate community structures in the French Rhône River. *Oecologia*, 151(3), 544–559. https://doi.org/10.1007/s00442-006-0655-1
- Death, R. G. (2008). The effect of floods on aquatic invertebrate communities. In J. Lancaster, & R. A. Briers (Eds.), Aquatic insects: Challenges to populations. Proceedings of the Royal Entemological Society's 24th Symposium (Chapter 6, pp. 103–121). Oxfordshire, England: CABI.
- Development Core Team, R. (2013). R: A language and environment for statistical computing. Vienna, Austria: R Foundation for Statistical Computing.
- Durance, I., & Ormerod, S. J. (2007). Climate change effects on upland stream macroinvertebrates over a 25-year period. *Global Change Biology*, *13*(5), 942–957. https://doi.org/10.1111/j.1365-2486.2007.01340.x
- Eastburn, D. (1990). The River. In N. Mackay & D. Eastburn (Eds.), *The Murray* (Chapter 1, pp. 3–16). Canberra, Australia: Murray Darling Basin Commission.
- Evans, R., Brown, C., & Kellett, J. (1990). Geology and groundwater. In N. Mackay & D. Eastburn (Eds.), *The Murray* (Chapter 5, pp. 77–94). Canberra, Australia: Murray Darling Basin Commission.
- Faith, D. P., Minchin, P. R., & Belbin, L. (1987). Compositional dissimilarity as a robust measure of ecological distance. Vegetatio, 69(1-3), 57–68. https://doi.org/10.1007/BF00038687
- Floury, M., Usseglio-Polatera, P., Ferreol, M., Delattre, C., & Souchon, Y. (2013). Global climate change in large European rivers: Long-term effects on macroinvertebrate communities and potential local confounding factors. *Global Change Biology*, *19*(4), 1085–1099. https://doi. org/10.1111/gcb.12124

Frontier, S. (1976). Étude de la décroissance des valeurs propres dans une analyse en composantes principales: Comparaison avec le module du bâton brisé. *Journal of Experimental Marine Biology and Ecology*, 25(1), 67–75. https://doi.org/10.1016/0022-0981(76)90076-9

Fruget, J.-F., Centofanti, M., Dessaix, J., Olivier, J.-M., Druart, J.-C., & Martinez, P.-J. (2001). Temporal and spatial dynamics in large rivers: Example of a long-term monitoring of the middle Rhône River. Annales de Limnologie - International Journal of Limnology, 37(3), 237–251. https://doi.org/10.1051/limn/2001021

Fruget, J.-F., Jézéquel, C., Archambaud, G., Dessaix, J., & Roger, M.-C. (2015). Long-term effects of global and local changes on benthic macroinvertebrate communities in multi-stressed large rivers: The example of the Rhône River during the last 30 years. *Knowledge and Management of Aquatic Ecosystems*, 416. https://doi.org/10.1051/kmae/2015025

Gauch, H. G. (1982). Noise reduction by eigenvector ordinations. Ecology, 63(6), 1643-1649. https://doi.org/10.2307/1940105

Gauch, H. G., Whittaker, T. R., & Wentworth, T. R. (1977). A comparative study of reciprocal averaging and other ordination techniques. Journal of Ecology, 65(1), 157–174.

Graca, M. A. S., Pinto, P., Cortes, R., Coimbra, N., Oliveira, S., Morais, M., et al. (2004). Factors affecting macroinvertebrate richness and diversity in Portuguese streams: A two-scale analysis. *International Review of Hydrobiology*, *89*(2), 151–164. https://doi.org/10.1002/ iroh.200310705

Hart, B. T., Bailey, P., Edwards, R., Hortle, K., James, K., McMahon, A., et al. (1991). A review of the salt sensitivity of the Australian freshwater biota. *Hydrobiologia*, 210(1-2), 105–144. https://doi.org/10.1007/BF00014327

Herczeg, A. L., Dogramaci, S. S., & Leaney, F. W. J. (2001). Origin of dissolved salts in a large, semi-arid groundwater system: Murray Basin, Australia. Marine and Freshwater Research, 52(1), 41–52. https://doi.org/10.1071/MF00040

Horrigan, N., Choy, S., Marshall, J., & Recknagel, F. (2005). Response of stream macroinvertebrates to changes in salinity and the development of a salinity index. *Marine and Freshwater Research*, 56(6), 825–833. https://doi.org/10.1071/MF04237

Hussain, Q. A., & Pandit, A. K. (2012). Macroinvertebrates in streams: A review of some ecological factors. International Journal of Fisheries and Aquaculture, 4(7), 114–123.

Isaak, D. J., & Rieman, B. E. (2013). Stream isotherm shifts from climate change and implications for distributions of ectothermic organisms. Global Change Biology, 19(3), 742–751. https://doi.org/10.1111/gcb.12073

Jackson, D. A. (1993). Stopping rules in principal components analysis: A comparison of heuristic and statistical approaches. *Ecology*, 74(8), 2204–2214. https://doi.org/10.2307/1939574

Jackson, J. K., & Füreder, L. (2006). Long-term studies of freshwater macroinvertebrates: A review of the frequency, duration and ecological significance. *Freshwater Biology*, *51*(3), 591–603. https://doi.org/10.1111/j.1365-2427.2006.01503.x

Jacobsen, T., & Adams, R. M. (1958). Salt and silt in ancient mesopotamian agriculture. Science, 128(3334), 1251–1258. https://doi.org/ 10.1126/science.128.3334.1251

Johnson, R. A., & Wichern, D. W. (2002). Applied multivariate statistical analysis, (5th ed.). Upper Saddle River, NY: Prentice-Hall, Inc.

Junk, W. J., Bayley, P. B., & Sparks, R. E. (1989). The flood pulse concept in river-floodplain systems. Canadian Special Publication of Fisheries and Aquatic Sciences, 106, 110–127.

Kefford, B. J., Marchant, R., Schäfer, R. B., Metzeling, L., Dunlop, J. E., Choye, S. C., et al. (2011). The definition of species richness used by species sensitivity distributions approximates observed effects of salinity on stream macroinvertebrates. *Environmental Pollution*, 159(1), 302–310. https://doi.org/10.1016/j.envpol.2010.08.025

Kefford, B. J., Nugegoda, D., Metzeling, L., & Fields, E. J. (2006). Validating species sensitivity distributions using salinity tolerance of riverine macroinvertebrates in the southern Murray-Darling Basin (Victoria, Australia). *Canadian Journal of Fisheries and Aquatic Sciences*, 63(8), 1865–1877. https://doi.org/10.1139/f06-080

Kefford, B. J., Papas, P. J., Metzeling, L., & Nugegoda, D. (2004). Do laboratory salinity tolerances of freshwater animals correspond with their field salinity? *Environmental Pollution*, 129(3), 355–362. https://doi.org/10.1016/j.envpol.2003.12.005

Krug, E. C., & Frink, C. R. (1983). Acid rain on acid soil: A new perspective. Science, 221(4610), 520–525. https://doi.org/10.1126/ science.221.4610.520

Lake, P. S. (2000). Disturbance, patchiness, and diversity in streams. Journal of the North American Benthological Society, 19(4), 573–592. https://doi.org/10.2307/1468118

Lawrence, B., & Paterson, L. (2005). *Review of the River Murray Water Quality Moniroing Program, MDBC Publication No. 15/05.* Canberra, Australia: Murray-Darling Basin Commission. Retrieved from https://www.mdba.gov.au/sites/default/files/archived/mdbc-tlm-reports/2111\_RM\_WQ\_monitoring\_program.pdf

Legendre, P., & Anderson, M. J. (1999). Distance-based redundancy analysis: Testing multispecies responses in multifactorial ecological experiments. *Ecological Monographs*, *69*(1), 1–24. https://doi.org/10.1890/0012-9615(1999)069[0001:DBRATM]2.0.CO;2

Legendre, P., & Legendre, L. (2012). Numerical ecology, (3rd English ed.). Amsterdam, The Netherlands: Elsevier.

Lepš, J., & Šmilauer, P. (2003). Multivariate analysis of ecological data using CANOCO. Cambridge, United Kingdom: Cambridge University Press. https://doi.org/10.1017/CBO9780511615146

Mackay, N., Hillman, T., & Rolls, J. (1988). Water quality of the Murray River: Review of monitoring 1978 to 1986. Canberra, Australia: Murray-Darling Basin Commission.

Macumber, P. (1990). The salinity problem. In N. Mackay & D. Eastburn (Eds.), *The Murray* Chapter 7, (pp. 111–126). Canberra, Australia: Murray Darling Basin Commission.

Madej, M., & Ozaki, V. (1996). Channel response to sediment wave propagation and movement, Redwood Creek, California, USA. *Earth Surface Processes and Landforms*, 21(10), 911–927. https://doi.org/10.1002/(SICI)1096-9837(199610)21:10<911::AID-ESP621>3.0. CO;2-1

Maheshwari, B. L., Walker, K. F., & McMahon, T. A. (1995). Effects of regulation on the flow regime of the River Murray, Australia. *Regulated Rivers: Research & Management*, 10(1), 15–38. https://doi.org/10.1002/rrr.3450100103

Mannes, S., Robinson, C. T., Uehlinger, U., Scheurer, T., Ortlepp, J., Mürle, U., et al. (2008). Ecological effects of a long-term flood program in a flow-regulated river. *Journal of Alpine Research*, 96(1), 125–134.

McKay, J. (2008). The legal frameworks of Australian water: Progression from Common Law rights to sustainable shares. In L. Crase (Ed.), Water policy in Australia: The impact of change and uncertainty (Chapter 4, pp. 44–60). Washington, DC: RFF Press.

McMahon, T. A., & Finlayson, B. L. (1995). Reservoir system management and environmental flows. Lakes & Reservoirs: Research and Management, 1(1), 65–76. https://doi.org/10.1111/j.1440-1770.1995.tb00007.x

Murray-Darling Basin Authority (2012). The River Murray system. Retrieved from https://riverdata.mdba.gov.au/system-view Murray-Darling Basin Authority (2014). *Basin-wide environmental watering strategy*. Canberra, Australia: Murray-Darling Basin Authority.

- MDBA (2015). Basin Salinity Management 2030: Status Report 2015–16. Canberra, Australia: Murray-Darling Basin Authority. Retrieved from https://www.mdba.gov.au/sites/default/files/pubs/D16-34851-basin-salinity-management-BSM2030-2015-16-status-report.pdf
- MDBA (2016). The Living Murray 2014-15 environmental watering report. Canberra, Australia: Murray-Darling Basin Authority. Retrieved from https://www.mdba.gov.au/publications/mdba-reports/living-murray-2014-15-environmental-watering-report

MDBA (2017a). Murray-Darling Basin. (07/17). Canberra, Australia: Murray-Darling Basin Authority. Retrieved from https://www.mdba.gov.au/ sites/default/files/pubs/696-revised-A0-Basin-map-web.pdf

MDBA (2017b). Progress on water recovery. Retrieved from https://www.mdba.gov.au/managing-water/environmental-water/progresswater-recovery

MDBC (2007). Living Murray business plan. Canberra, Australia: Murray-Darling Basin Commission.

MDBMC (1994). The algal management strategy for the Murray-Darling Basin. Canberra, Australia: Murray-Darling Basin Ministerial Council. Retrieved from https://www.mdba.gov.au/sites/default/files/pubs/the-algal-management-strategy.pdf

MDBMC (2015). Basin Salinity Management 2030. Canberra, Australia: Murray-Darling Basin Ministerial Council. Retrieved from https://www. mdba.gov.au/publications/mdba-reports/basin-salinity-management-2030

MDFRC (2013). Identification and ecology of Australian freshwater invertebrates. The Murray-Darling Freshwater Research Centre (MDFRC). Retrieved from http://www.mdfrc.org.au/bugguide/index.htm

Mills, B. J., & Geddes, M. C. (1980). Salinity tolerance and osmoregulation of the Australian freshwater crayfish Cherax destructor Clark (Decapoda: Parastacidae). Australian Journal of Marine and Freshwater Research, 31(5), 667–676. https://doi.org/10.1071/MF9800667

Mosley, L. (2015). Drought impacts on the water quality of freshwater systems; review and integration. *Earth-Science Reviews*, 140, 203–214. https://doi.org/10.1016/j.earscirey.2014.11.010

Musgrave, W. (2008). Historical development of water resources in Australia: Irrigation policy in the Murray-Darling Basin. In L. Crase (Ed.), Water policy in Australia: The impact of change and uncertainty (Chapter 3, pp. 28–43). Washington, DC: RFF Press.

Neave, I., McLeod, A., Raisin, G., & Swirepik, J. (2015). Managing water in the Murray-Darling Basin under a variable and changing climate. WATER, April, 102–107. Retrieved from https://www.mdba.gov.au/sites/default/files/pubs/Managing%20water%20in%20the%20murraydarling%20basin%20under%20a%20variable%20and%20changing%20climate.pdf

Nilsson, C., Reidy, C. A., Dynesius, M., & Revenga, C. (2005). Fragmentation and flow regulation of the world's large river systems. Science, 308(5720), 405–408. https://doi.org/10.1126/science.1107887

Økland, R. H. (1999). On the variation explained by ordination and constrained ordination axes. *Journal of Vegetation Science*, 10(1), 131–136. https://doi.org/10.2307/3237168

Oksanen, J., Blanchet, F. G., Kindt, R., Legendre, P., Minchin, P. R., O'Hara, B., et al. (2013). Vegan: Community ecology package (version R package version 2.0-7). Retrieved from http://CRAN.R-project.org/package=vegan

Oksanen, J., & Minchin, P. R. (2002). Continuum theory revisited: What shape are species responses along ecological gradients? *Ecological Modelling*, 157(2-3), 119–129. https://doi.org/10.1016/S0304-3800(02)00190-4

Pagan, P. (2008). Adaptive management. In L. Crase (Ed.), Water policy in Australia: The impact of change and uncertainty (Chapter 14, pp. 216–229). Washington, DC: RFF Press.

Paul, W. L., & Anderson, M. J. (2013). Casual modelling with multivariate species data. Journal of Experimental Marine Biology and Ecology, 448, 72–84. https://doi.org/10.1016/j.jembe.2013.05.028

Paul, W. L., Cook, R. A., Suter, P. J., Clarke, K. R., Shackleton, M. E., McInerney, P. J., et al. (2018). Murray River Water Quality Monitoring Program (WQMP) macroinvertebrate and environmental data. https://doi.org/10.4225/22/5aa5ec0fde035

Pearl, J. (2000). Causality: Models, reasoning, and inference (pp. 201–258). New York, NY: Cambridge University Press.

Pearl, J. (2014). The causal foundations of structural equation modeling. In R. H. Hoyle (Ed.), Handbook of structural equation modeling (Chapter 5, pp. 68–91). New York, NY: Guilford Press.

Pillsbury, A. F. (1981). The salinity of rivers. Scientific American, 245(1), 54–65. https://doi.org/10.1038/scientificamerican0781-54

Quinn, J. M., Cooper, A. B., Davies-Colley, R. J., Rutherford, J. C., & Williamson, R. B. (1997). Land use effects on habitat, water quality, periphyton, and benthic invertebrates in Waikato, New Zealand, hill-country streams. New Zealand Journal of Marine and Freshwater Research, 31(5), 579–597. https://doi.org/10.1080/00288330.1997.9516791

Reice, S. R., Wissmar, R. C., & Naiman, R. J. (1990). Disturbance regimes, resilience, and recovery of animal communities and habitats in lotic ecosystems. *Environmental Management*, 14(5), 647–659. https://doi.org/10.1007/BF02394715

Resh, V. H., Brown, A. V., Covich, A. P., Gurtz, M. E., Li, H. W., Minshall, G. W., et al. (1988). The role of disturbance in stream ecology. Journal of the North American Benthological Society, 7(4), 433–455. https://doi.org/10.2307/1467300

Reynoldson, T. B., Norris, R. H., Resh, V. H., Day, K. E., & Rosenberg, D. M. (1997). The reference condition: A comparison of multimetric and multivariate approaches to assess water-quality impairment using benthic macroinvertebrates. *Journal of the North American Benthological Society*, 16(4), 833–852. https://doi.org/10.2307/1468175

Rier, S. T., & Stevenson, R. J. (2006). Response of periphytic algae to gradients in nitrogen and phosphorus in streamside mesocosms. *Hydrobiologia*, 561(1), 131–147. https://doi.org/10.1007/s10750-005-1610-6

Robinson, C. T. (2012). Long-term changes in community assembly, resistance, and resilience following experimental floods. *Ecological Applications*, 22(7), 1949–1961. https://doi.org/10.1890/11-1042.1

Rutherford, J. C., & Kefford, B. J. (2005). Effects of salinity on stream ecosystems: Improving models for macroinvertebrates, CSIRO Land and Water Tech. Rep. 22/05, (). Canberra, Australia: CSIRO Land and Water.

Schäfer, R. B., Kefford, B. J., Metzeling, L., Liess, M., Burgert, S., Marchant, R., et al. (2011). A trait database of stream invertebrates for the ecological risk assessment of single and combined effects of salinity and pesticides in south-East Australia. *Science of the Total Environment*, 409(11), 2055–2063. https://doi.org/10.1016/j.scitotenv.2011.01.053

Scheffer, M., Carpenter, S., Foley, J. A., Folke, C., & Walker, B. (2001). Catastrophic shifts in ecosystems. *Nature*, 413(6856), 591–596. https://doi. org/10.1038/35098000

Schrobback, P., Adamson, D., & Quiggin, J. (2008). Options for salinity mitigation in the Murray-Darling Basin. Retrieved from The University of Queensland: https://rsmq.group.uq.edu.au/files/811/WPM08\_1.pdf

- Scott, B. J., Fenton, I. G., Fanning, A. G., Schumann, W. G., & Castleman, L. J. C. (2007). Surface soil acidity and fertility in the eastern Riverina and Western Slopes of southern New South Wales. *Australian Journal of Experimental Agriculture*, 47(8), 949–964. https://doi.org/10.1071/ EA05155x
- Scrimgeour, G. J., & Winterbourn, M. J. (1989). Effects of floods on epilithon and benthic macroinvertebrate populations in an unstable New Zealand river. *Hydrobiologia*, 171(1), 33–44. https://doi.org/10.1007/BF00005722

Smedema, L. K., & Shiati, K. (2002). Irrigation and salinity: A perspective review of the salinity hazards of irrigation development in the arid zone. Irrigation and Drainage Systems, 16(2), 161–174. https://doi.org/10.1023/A:1016008417327 Smith, D. I. (1998). Water in Australia: Resources and management. Melbourne, Australia: Oxford University Press Australia.

Somerfield, P. J., & Clarke, K. R. (2013). Inverse analysis in non-parametric multivariate analyses: Distinguishing groups of associated species which covary coherently across samples. *Journal of Experimental Marine Biology and Ecology*, 449, 261–273. https://doi.org/10.1016/j. jembe.2013.10.002

ter Braak, C. J. F. (1985). Correspondence analysis of incidence and abundance data: Properties in terms of a unimodal response model. Biometrics, 41(4), 859–873. https://doi.org/10.2307/2530959

ter Braak, C. J. F., & Prentice, I. C. (1988). A theory of gradient analysis. Advances in Ecological Research, 18, 271–317. https://doi.org/10.1016/ S0065-2504(08)60183-X

ter Braak, C. J. F., & Šmilauer, P. (2002). CANOCO reference manual and CanoDraw for Windows user's guide: Software for canonical community ordination (version 4.5). Ithaca, NY: Microcomputer Power.

Thoms, M., Suter, P., Roberts, J., Koehn, J., Jones, G., Hillman, T., et al. (2000). Report of the River Murray scientific panel on environmental flows. River Murray—Dartmouth to Wellington and the Lower Darling River. Canberra, Australia: Murray-Darling Basin Commission.

Timbal, B., Abbs, D., Bhend, J., Chiew, F., Church, J., Ekström, M., et al. (2015). Murray Basin cluster report. In M. Ekström, P. Whetton, C. Gerbing, M. Grose, L. Webb, & J. Risbey (Eds.), Climate change in Australia projections for Australia's Natural Resource Management Regions: Cluster reports. Australia: CSIRO and Bureau of Meteorology. Retrieved from https://www.climatechangeinaustralia.gov.au/media/ ccia/2.1.6/cms\_page\_media/172/MURRAY\_BASIN\_CLUSTER\_REPORT\_1.pdf

Townsend, C. R., & Scarsbrook, M. R. (1997). The intermediate disturbance hypothesis, refugia, and biodiversity in streams. *Limnology and Oceanography*, 42(5), 938–949. https://doi.org/10.4319/lo.1997.42.5.0938

Tripole, S., Vallania, E., & Corigliano, M. (2008). Benthic macroinvertebrate tolerance to water acidity in the Grande river sub-basin (San Luis, Argentina). *Limnetica*, 27(1), 29–38.

Vannote, R. L., Minshall, G. W., Cummins, K. W., Sedell, J. R., & Cushing, C. E. (1980). The river continuum concept. Canadian Journal of Fisheries and Aquatic Sciences, 37(1), 130–137. https://doi.org/10.1139/f80-017

Vaughan, I. P., & Ormerod, S. J. (2012). Large-scale, long-term trends in British river macroinvertebrates. Global Change Biology, 18(7), 2184–2194. https://doi.org/10.1111/j.1365-2486.2012.02662.x

Ward, J. V., & Stanford, J. A. (1983). The serial discontinuity concept of lotic ecosystems. In T. D. Fontaine & S. M. Bartlell (Eds.), Dynamics of Lotic Ecosystems (Chapter 2, pp. 29–42). Ann Arbor, MI: Ann Arbor Science.

Water Act 2007 - Basin Plan (2012). Retrieved from https://www.legislation.gov.au/Details/F2012L02240

Wood, S. N. (2006). Generalized additive models: An introduction with R. Boca Raton, FL: Chapman and Hall. https://doi.org/10.1201/ 9781420010404